



# Article EU Net-Zero Policy Achievement Assessment in Selected Members through Automated Forecasting Algorithms

Cristiana Tudor <sup>1</sup>,\*<sup>1</sup> and Robert Sova <sup>2</sup>

- <sup>1</sup> International Business and Economics Department, Bucharest University of Economic Studies, 010374 Bucharest, Romania
- <sup>2</sup> Department of Management Information Systems, The Bucharest University of Economic Studies, 010374 Bucharest, Romania; robert.sova@ase.ro
- \* Correspondence: cristiana.tudor@net.ase.ro

Abstract: The European Union (EU) has positioned itself as a frontrunner in the worldwide battle against climate change and has set increasingly ambitious pollution mitigation targets for its members. The burden is heavier for the more vulnerable economies in Central and Eastern Europe (CEE), who must juggle meeting strict greenhouse gas emission (GHG) reduction goals, significant fossil-fuel reliance, and pressure to respond to current pandemic concerns that require an increasing share of limited public resources, while facing severe repercussions for non-compliance. Thus, the main goals of this research are: (i) to generate reliable aggregate GHG projections for CEE countries; (ii) to assess whether these economies are on track to meet their binding pollution reduction targets; (iii) to pin-point countries where more in-depth analysis using spatial inventories of GHGs at a finer resolution is further needed to uncover specific areas that should be targeted by additional measures; and (iv) to perform geo-spatial analysis for the most at-risk country, Poland. Seven statistical and machine-learning models are fitted through automated forecasting algorithms to predict the aggregate GHGs in nine CEE countries for the 2019-2050 horizon. Estimations show that CEE countries (except Romania and Bulgaria) will not meet the set pollution reduction targets for 2030 and will unanimously miss the 2050 carbon neutrality target without resorting to carbon credits or offsets. Austria and Slovenia are the least likely to meet the 2030 emissions reduction targets, whereas Poland (in absolute terms) and Slovenia (in relative terms) are the farthest from meeting the EU's 2050 net-zero policy targets. The findings thus stress the need for additional measures that go beyond the status quo, particularly in Poland, Austria, and Slovenia. Geospatial analysis for Poland uncovers that Krakow is the city where pollution is the most concentrated with several air pollutants surpassing EU standards. Short-term projections of PM2.5 levels indicate that the air quality in Krakow will remain below EU and WHO standards, highlighting the urgency of policy interventions. Further geospatial data analysis can provide valuable insights into other geo-locations that require the most additional efforts, thereby, assisting in the achievement of EU climate goals with targeted measures and minimum socio-economic costs. The study concludes that statistical and geo-spatial data, and consequently research based on these data, complement and enhance each other. An integrated framework would consequently support sustainable development through bettering policy and decision-making processes.

**Keywords:** automated forecasting; GHG emissions; European Green Deal; neural network autoregression model (NNAR); statistical methods; aggregated data

# 1. Introduction

Climate change and environmental degradation are currently major concerns for Europe and the rest of the globe [1]. Climate change is considered a serious problem by 93% of Europeans, and almost eight out of ten EU individuals (78%) believe it is an extremely serious issue [2]. In consideration of the above, the European Commission (EC) unveiled



Citation: Tudor, C.; Sova, R. EU Net-Zero Policy Achievement Assessment in Selected Members through Automated Forecasting Algorithms. *ISPRS Int. J. Geo-Inf.* **2022**, *11*, 232. https://doi.org/ 10.3390/ijgi11040232

Academic Editors: Joep Crompvoets, Cesar Casiano Flores and Wolfgang Kainz

Received: 12 February 2022 Accepted: 25 March 2022 Published: 31 March 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). in December 2019 the ambitious European Green Deal [3,4], a package of policy measures spearheaded by the EC with the goal of achieving climate neutrality in the European Union (EU) by 2050 and an intermediate target of cutting carbon emissions by at least 50 percent (and toward 55 percent) relative to 1990 levels by 2030 [5].

Moreover, the pledge toward the green transition has recently been acknowledged as one of the key elements in addressing the economic meltdown caused by the COVID-19 pandemic [6,7] and thus regarded as the lifeline out of the COVID-19 pandemic [1]. Consequently, one-third of the 1.8 trillion Euro investments within the European Recovery Fund announced in July 2020 will finance the European Green Deal's objectives [1,8].

The European Council, comprising the heads of the EU's Member States, further approved the new binding EU objective of at least 55% reduction in GHG emissions by 2030 in December 2020 [8]. Finally, the European Climate Law adopted in June 2021 enforced the European Green Deal objective of making Europe's economy and society climate-neutral by 2050, including its intermediate 2030 goal to reduce net greenhouse gas emissions by at least 55 percent from 1990 levels [9–11].

As such, Europe has long positioned itself as the frontrunner in the global fight against climate change and is making a successful transition to a low-emissions economy [12]. Figure 1 confirms that the EU managed to reduce its greenhouse gas emissions by 23% between 1970 and 2018, whereas the world average registered an increasing trend (i.e., a 69.54% growth) over the same period.



**Figure 1.** Historical evolution of GHGs over 1970–2018. Authors' representation. Data source: World Development Indicators (WDI).

While under the global reach of the Paris Agreement [13,14], individual countries' pledges to reduce their total emissions (i.e., National Determined Contributions, or NDCs) are not legally binding or enforceable [15], the emission reduction targets under the EU Green Deal are mandatory for EU member states [16]. This could, by itself, pose significant problems for Europe's more fragile economies in the CEE area that have not yet been able to catch up with their western counterparts [17–19]. Moreover, the COVID-19 induced

recession is expected to be significantly more severe in smaller CEE economies compared with in the more developed EU states [20], which further adds to the pressure of meeting binding EU climate targets while dealing with the effects of the largest health and economic crisis in recent history [21] and managing post-pandemic recovery.

Although this is a timely issue worldwide, as all countries have to balance postpandemic recovery and polluting emissions mitigation [22], the CEE economies also deal with the added pressure of legally binding reduction targets. As a result, their implementation may on one hand pose regulatory challenges [23], whereas, on the other hand, noncompliance could have serious repercussions, including infringement procedures, legal challenges, and ultimately financial penalties [24].

Additionally, most governments in the CEE area remain reluctant to implement ambitious pollution reduction policies set by the European Commission, mainly because their economies are still significantly dependent on coal and other fossil fuels. For example, as of 2018, the highest shares of coal in power generation were found in Poland (80%), the Czech Republic (54%), and in Bulgaria (43%) [25].

Poland employs roughly half of the coal workforce at the EU level, followed by Germany, the Czech Republic, Romania, Bulgaria, Greece, and Spain [26], which further complicates the task of implementing the European Green Deal and meeting emission reduction targets. Additionally, citizens in Central and Eastern Europe are generally less concerned about global warming than those in Western Europe [27–30], which also contributes to spurring CEE government reluctance to move toward carbon neutrality.

Although polluting emissions in CEE fell after the fall of the Berlin Wall and the dissolution of the Soviet Union, and the area is currently well behind the world average when it comes to the contribution to world pollution (Figure 2), it should be nonetheless recognized that this is mainly a product of diplomatic compromise and the collapse of an inefficient communist industry, rather than the result of national pollution reduction policies [25], and that the reduction trends remain insufficient compared with the new net 55% reduction target for 2030.



**Figure 2.** Worldwide GHG emissions in 2018. Authors' representation. Source of data: World Development Indicators (WDI).

The formulation of pollution-mitigation policies, as well as its monitoring, intrinsically relies on emissions estimates. Thus, reliable estimates of emission trends are essential for sound policy-making, and national and international bodies are increasingly using forecasts of polluting emissions in their policy-making processes [31]. This, in turn, is a major motivating factor for this research. Furthermore, accurately forecasting emissions and reversing rising trends early-on is especially critical for the more vulnerable EU members, preventing significant costs in the form of infringement procedures, legal battles, and, ultimately, financial penalties. Hence, this study further contributes to assessing

whether the CEE countries are on the right path toward the achievement of their binding pollution mitigation targets, under the assumption that the current conditions and measures remain unchanged.

Consequently, we identify the countries that are farthest from meeting their pollution reduction targets and thus highlight where future research using spatial inventories of GHGs at a finer resolution is needed to uncover specific areas that should be targeted by additional climate combat measures. The results of this research are, on one hand, critical for policy monitoring and its potential revision at the EU level by considering the particularities of the CEE economies and their specific difficulties, and on the other hand for national governments and agencies for early-warning purposes. Overall, we scientifically reason that synergy between EU policy and national measures is key for reaching climate combat goals at the EU level. In turn, GHG emissions forecasts serve as the foundation.

Producing accurate forecasts of polluting emissions remains a challenging research task [32]. Mounting empirical evidence recognizes polluting emissions as a leading factor for socio-economic indicators, including economic growth, mortality, and health variables [33–36]. The univariate time series forecasting allows producing evidence for a leading factor unaffected by other variables and brings further advantages in the form of increased reliability and reduced risk of model misspecification [37].

However, the strand of literature on forecasting polluting emissions over the long term within a univariate framework, particularly with a focus on Central and Eastern Europe, remains thin. This study contributes to filling the void by providing new insights on future emissions trends for a relevant sample of nine CEE countries (i.e., Austria, Bulgaria, Croatia, Czech Republic, Hungary, Poland, Romania, Slovak Republic, and Slovenia).

The univariate methods for forecasting air quality that have been proposed in the extant literature belong to the two major categories delineated by [38], namely statistical methods and deep-learning methods [39]. Ref. [40] used the logistic equation to model  $CO_2$  emissions in three Chinese industries and further combined estimations to explain the total  $CO_2$  emissions in China.

Ref. [41] employed several econometric methods, including the ARIMA model, Holt–Winters, exponential smoothing, and singular spectrum analysis (SSA). They also introduced a new combination forecast, which includes the SSA and the forecast provided by the Energy Information Administration (EIA). The new combination model was found to out-perform the other candidates and was further used to provide 12-month ahead forecasts for US  $CO_2$  emissions energy series.

More recently, [42] used a univariate autoregressive integrated moving average (ARIMA) and several scenarios to forecast  $CO_2$  emissions for Pakistan for the 2030 horizon. However, this excludes alternative forecasting methods. Authors that employ machine-learning models generally confirm their suitability for predicting polluting emissions and conclude their forecasting superiority relative to statistical methods.

Ref. [43] used statistical and machine-learning models, including a naive model, ETS, STS, TBATS, HW, ARIMA, and the neural network autoregressive model (NNAR) for modeling and forecasting the evolution of  $CO_2$  emissions in Bahrain. They reported that the NNAR model outperformed in the out-of-sample setting model. More recently, [44] modeled and forecast the trend of Bahrain's  $CO_2$  emissions by employing multiple forecasting methods (i.e., the Gaussian process regression rational quadratic model, the neural network time series nonlinear autoregressive model, and Holt's method. They also identified the NNAR model as outperforming the other methods by showing the lowest root mean square errors (RMSE) for out-of-sample predictions.

Finally, [45] offered a more extensive study on forecasting polluting emissions. They employed seven statistical and machine-learning methods that are also used in our investigation to predict the GHG emission trends in the twelve world's top polluters. Similar to its predecessors, it also establishes that NNAR is superior in terms of predictive ability at different forecasting horizons by reporting both the lowest root mean square error (RMSE)) and mean absolute scaled error (MASE). Of note, the vast majority of previous research

is usually focused on a single country and stops at exploring the predictive ability of concurrent methods by estimating forecasting accuracy metrics.

This study performs multi-country analysis and further defends its results against the Kolmogorov–Smirnov (KS) predictive accuracy test (KSPA). Additionally, we go further than previous research by monitoring EU net-zero policy implementation within the sample of CEE countries. This is achieved through estimations of GHGs policy-target values that are compared to GHGs estimations produced by the country-specific overperforming forecasting model. A preliminary assessment of the forecasting ability observed for the years 2019 and 2020 is also provided, indicating high accuracy (i.e., 98.5%) in predicting GHG emissions values for the year 2019, albeit, as expected, an overestimation of GHGs values for the year 2020, when the accuracy decreased to 90%, but remained satisfactory, especially in light of the black-swan event.

Furthermore, whereas most related studies employ CO2 emissions as the variable of interest, the current research inputs the more relevant variable of aggregate GHG emissions at a national level. Thus, while acknowledging that estimates of GHG emissions at a national level are less helpful for the development of pollution mitigation strategies and measures [46], we reason that accurate forecasting of aggregate GHG trends still has merits, as it is informative for policymaking purposes.

To this end, six predictive models are implemented through automated forecasting algorithms: the exponential smoothing state-space model (ETS), the Holt–Winters model (HW), the trigonometric ETS state-space model with Box–Cox transformation, ARMA errors, trend, and seasonal components (TBATS) model, the autoregressive integrated moving average (ARIMA) model, and the structural time series (STS) model) within the statistical methods category, and the neural network autoregression model (NNAR) within the machine-learning category.

A naive model, which always predicts the last observed value as per the usual approach in the time series forecasting literature, is also estimated, serving for comparative purposes. The results indicate that the neural network autoregression model (NNAR) shows the best out-of-sample forecasting performance for aggregate GHG emissions at the national level in the sample of CEE countries.

Furthermore, the research findings indicate, based on existing conditions, that CEE countries are projected to miss the mandatory reduction targets under the European Green Deal. Austria and Slovenia are farthest from meeting the 2030's 55% emissions reduction target, whereas Poland (in absolute terms) and Slovenia (in relative terms) are farthest from meeting the EU's 2050 net-zero policy target.

Consequently, the EU's post-COVID recovery plan constitutes a timely occasion to formulate and implement additional measures aimed at putting the CEE countries on the right track toward carbon neutrality. In this respect, this study opens another research direction where the proposed method can be replicated for specific cases on more reliable and finer-scale GHG emission inventories to reveal trouble areas and further intervene with targeted policies and measures, which would minimize social and economic costs for the still fossil-fuel-dependent CEE economies.

For exemplificative purposes, geo-spatial analysis is implemented for the country that is most at risk from meeting EU-set pollution mitigation targets, i.e., Poland. Our findings indicate that pollutants are highly concentrated in and around the city of Krakow, thus, pinpointing where targeted policy intervention should be directed. Forecasts of PM2.5 levels in Krakow indicate an upward trend that is a cause for alarm and highlights that, absent of further policy interventions, the air quality in Krakow will remain under EU standards with negative public health and economic consequences.

For other vulnerable CEE countries, geo-spatial inventories are not publicly available to our knowledge. Nonetheless, among others, [47–49] provided approaches for disaggregating national GHG data that can be further used to perform in-depth analyses for other countries that are projected to significantly miss policy targets under the status-quo hypoth-

esis. This research thus serves to provide a base for future research aimed to complement its findings.

Summing up the above consideration, the main goals of this research are (i) to generate reliable aggregate GHG projections for CEE countries; (ii) to assess whether these economies are on track to meet their binding pollution reduction targets; (iii) to pin-point the countries for which more in-depth analysis using spatial inventories of GHGs at a finer resolution is needed to uncover specific areas that should be targeted by additional measures; and (iv) to perform geo-spatial analysis for the most at-risk country, Poland.

The remainder of the paper is organized as follows. Section 2 explains the data and method employed in the empirical investigation. Further, Section 3 presents the empirical results, performs robustness checks, and provides estimates of emissions trends. Next, Section 4 discusses the main findings, whereas Section 5 concludes the study.

#### 2. Materials and Methods

### 2.1. Data

This study uses annual data on aggregate GHG emissions measured in kt of CO<sub>2</sub> covering the 31 December 1970–31 December 2018 period, or a total of 49 years, retrieved from the World Development Indicators (WDI) database of the World Bank. The time span for the dataset covers the maximum period of data availability. First, GHG emissions data were extracted from the WDI for all world countries with 49 available annual observations, resulting in a sample of 218 individual time series. Subsequently, data corresponding to nine Central and Eastern European countries that make up the subject of the current research (i.e., Austria, Bulgaria, Croatia, Czech Republic, Hungary, Poland, Romania, Slovak Republic, and Slovenia) was subset. Thus, the final dataset of the study comprises 49 annual observations for nine GHG emissions time series.

The following subsection provides an overview of the state of affairs in the nine CEE countries included in the analysis.

#### GHG Emissions in Selected Central and Eastern European Countries

Table 1 reflects the total GHG emissions registered during the most recent year of available data, i.e., 2018 in the nine CEE countries, together with the GHG emissions percentage changes over the past decades and policy-relevant periods. The average values at the European Union level as well as at the world level are also reported for comparative purposes.

**GHG Emissions** % Change % Change % Change Country 2018 (Total, kt **Income Category** (Relative to 2015) (Relative to 1990) (Relative to 1970) of CO<sub>2</sub>) Austria 74,980 -0.66-1.689.6 High income -6.94-45.77Bulgaria Upper middle 53,330 -39.42-0.62-12.4Croatia High income 22,550 -21.4High income 122,840 Czech Republic 0.45 -32.63-42.260,920 4.44 Hungary High income -31.64-30.56Poland High income 389,650 6.39 -11.91-16.42Romania Upper middle 109.010 0.85 -55.54-38.16Slovak Republic High income 39,930 5.72 -40.39-25.37Slovenia 17,170 5.27 High income -3.3254.52 European Union Aggregates 3,567,090 -1.50-21.69-22.38World 69.54 Aggregates 45,873,850 3.27 53.69 1.65 -27.14-15.60Average

Table 1. Greenhouse gas emissions in nine EE countries: 2018 data and historical evolution.

Of note, the CEE countries reflected in Table 1 mostly belong to the high-income category as per the World Bank classification, and only Romania and Bulgaria are uppermiddle-income economies. As of 2018, the top GHG emitters in the CEE area in absolute terms are Poland, the Czech Republic, and Romania.

Together, the three countries account for almost 70% of 2018 GHG emissions in CEE, with Poland alone responsible for 43.76% of GHG emissions in the area. The vast majority of CEE countries recorded reductions in GHG emissions between 1970 and 2018, with the Czech Republic leading the way (-42.2% between 1970 and 2018), followed by Bulgaria (-39.42%), and Romania (-38.16%). The GHG decrease in these countries is much greater when compared to the 1990 baseline—a development that is directly tied to the demise of inefficient communist industries. Only two CEE countries have shown increases in polluting emissions since 1970, with Slovenia reporting the highest growth rate (54.52%) and Austria reporting a far more modest rate of 9.6%. Over the same time period, at the EU level, GHG emissions decreased on average by 22.38%, whereas the nine CEE countries reported an average reduction of 15.60%.

However, they did manage to surpass the EU average in the aftermath of the communist regime collapse (a decrease of 27.14% for the CEE countries relative to an average decrease of 21.69% at the EU level). Although, over the entire 1970–2018 period, the overall emissions in CEE countries exhibited a declining trend, they did not manage to match their more developed counterparts within the EU.

However, there are disparities in the emission trends at the CEE level, as further reflected in Figure 3, showing that emissions in Poland, Romania, and the Czech Republic remained well above the levels registered in other CEE countries, despite their more accelerated decrease over the past decades.





**Figure 3.** Total GHG emission trends in CEE countries. Authors' representation. Source of data: World Development Indicators (WDI).

#### 2.2. Method

An overview of the research method implemented to offer an assessment of the EU climate-policy implementation progress in the nine Central and Eastern European countries is first depicted in Figure 4, whereas more details on its main building blocks are provided in the following sub-sections.



Figure 4. Research methodology flowchart.

The research method is based on two crucial elements: (i) the forecasting technique employed to produce forecasts and (ii) the forecasting models encompassed into the algorithm for automated forecasting purposes

### 2.2.1. The Holdout Period Forecasting

For forecasting purposes, the historical data series of length  $N_i$  ( $i \in \{1, ..., 9\}$  and N = 49) is separated into two subsets corresponding to a training (or fit) period and a test period or lead-time period to conduct the holdout method [50]. The data up to 2010 (about 84 percent of observations) are employed in-sample for model training and validation, while the data from 2011 to 2018 (roughly 16 percent of observations) are used to assess the predictive models' out-of-sample forecasting accuracy.

As the two most important benchmarks specified within the European Green Deal for emission reduction targets are the years 2030 and 2050, we are particularly interested in finding the best forecasting model within the universe of seven candidates over the lead-time and subsequently using it for providing h steps ahead forecasts corresponding to the two forecasting horizons for GHGs in the nine CEE countries. Thus, h is set to 32 (so as

to cover the 2019–2050 forecasting horizon), and the point forecasts for years 2030 and 2050 are highlighted. Figure 5 gives a graphical representation of the holdout cross-validation method [51] employed in this research.





2.2.2. The Automated Forecasting Models

Machine-learning methods and traditional statistical methods are implemented through automated forecasting algorithms to predict the GHG trends. Figure 6 depicts the univariate forecasting models employed in this study for automated forecasting, delineated in the two categories proposed by [38].



Figure 6. The forecasting models employed for automated forecasting.

Hence, the predictive models are automatically implemented in R software through specific functions mainly included in two main packages, the "stats" package [52] and the "forecast" package [53], respectively.

The naive model specified in Equation (1) is manually coded in R such that the k-step-ahead forecast equals the observed GHG emissions value for country i at time t, such as

$$F_i(t+k) = y_i(t) \tag{1}$$

A great advantage of the automatization of model estimation is that a specified fitness function (i.e., the AIC, AICc, and BIC) is run to search over a feature space (pool of covariates) to select the optimal model parameters, thus, assuring that the in-sample fit is maximized. Consequently, the algorithm searches through multiple pools of covariates to identify the optimal parameters for 63 models (i.e., 7 models × 9 countries, or  $M_{ij}$ , i = 1,...,9, j == 1, ..., 7) and thus identifies parameters of the seven models that assure maximum fit on training data for each country.

The single best-performing model for each country further emerges through the estimation of various accuracy metrics for the out-of-sample predictive ability. Its forecasting superiority is subsequently assessed through formal SPA testing. Further details on implementing traditional statistical methods in the R environment are provided next, while more information can be found in [54,55].

The exponential smoothing state-space (ETS) model was developed by [56] through extending the standard Exponential Smoothing (ES) approach. A forecast equation and a smoothing equation are included in the basic ETS model, which is then incorporated into an innovation state-space model. Each time series is handled as a mixture of three components: the trend (T), seasonal (S), and error (E) components in exponential smoothing, with the trend component further being a combination of a level term (l) and a growth term (b).

The trend and seasonal elements can be none (N), additive (A), additive damped (Ad), multiplicative (M), or multiplicative damped (Md) The final model is a three-character string (Z,Z,Z), with the first letter indicating the state-space model's error assumption, the second indicating the trend type, and the third indicating the season type [53,56,57]). The estimation of the ETS model is entirely automated using R's "forecast" package and its embedded "ets" function. In this research, the system is prompted to choose the error, type, and season autonomously and to identify the optimal parameters by running the corrected Akaike information criterion (AICc) as the fitness function.

*The Holt–Winters Model (HW)* was first proposed by [58,59] and is often referred to as double exponential smoothing. The model smooths the time series with three exponential smoothing formulae that are applied to the mean, trend, and each seasonal sub-series.

The exponential component  $(E_t)$  is given by

$$E_t = wY_t + (1 - w)(E_{t-1} + T_{t-1}), 0 < w < 1$$
<sup>(2)</sup>

The trend component is given by

$$T_t = v(E_t - E_{t-1}) + (1 - v)T_{t-1}, 0 < v < 1$$
(3)

Finally, the k-step-ahead forecast is issued as

$$F_{t+k|t} = E_t + kT_t \tag{4}$$

The "HoltWinters" function in the "stats" package in R software is used to automate the estimation of the HW model for the nine annual GHGs time series of interest by performing HW filtering on each series and using the squared prediction error to detect specific parameters.

The TBATS Model (Exponential Smoothing State Space Model with Box–Cox Transformation, ARMA Errors, Trend, and Seasonal Components) was developed by [60]. It can be automatically estimated in R through the "TBATS" function in the "forecast" package. The fitted model is specified as TBATS(*omega*, *phi*, *m*1, *k*1 >, ..., *m*J, *k*J >), where *omega* is the Box–Cox parameter; *phi* is the damping parameter; *m*1, ..., *m*J are the seasonal periods; and *k*1, ..., *k*J are the number of Fourier terms utilized for each seasonality. A TBATS model thus requires the estimation of 2(k1 + k2 + ..., kT) initial seasonal values. In our estimations, we instruct the algorithm to employ AIC to automatically identify the model parameters for each of the nine time series (countries). The ARIMA model (Autoregressive Integrated Moving Average) of [61] is a generalization of the ARMA (autoregressive moving average) model, widely popularized by [62]. A seasonal model is expressed as ARIMA (p,q,d)(P,Q,D)s, where s signifies the seasonal period and the lowercase and capital letters reflect the number of nonseasonal and seasonal parameters for each of its components, as per [54]. An ARIMA(p,d,q)(P,D,Q)s model is expressed in equation form as:

$$(1 - \varphi_1 B - \dots - \varphi_p B^p)(1 - \Phi_1 B^s - \dots - \Phi_p B^{s^p})(1 - B)^d (1 - B^s)^D Y_t = (1 - \theta_1 B - \dots - \theta_q B^q)(1 - \Theta_1 B^s - \dots - \Theta_p B^{sQ})\varepsilon_t$$
(5)

where  $\varepsilon_t$  is a random variable that has a mean of zero and the standard deviation  $\sigma$ .

The "auto.arima function" within R's software's "forecast" package is run to return the best ARIMA model for each series, by using unit root tests, AICc minimization, and MLE. The function can determine whether the data that are employed to fit the models require seasonal differencing and estimates unit root tests, while minimizing the AICc and MLE, to select parameters in a step-wise manner. As a result, instead of examining every potential combination of p and q, the "auto.arima" function yields substantially greater efficiency.

Structural Time Series Models (STS), developed by [63] are expressed explicitly in terms of non-obvious components, such as trends, cycles, and seasonals, which have a natural interpretation and represent the key elements of the series under study. The primary concept behind structural time series models is that they are built up as regression models with time-dependent explanatory variables and coefficients that change over time [64]. Equation (6) gives a generalized expression for the decomposition of a time series, such that

$$y_t = \mu_t + \psi_t + \gamma_t + \varepsilon_t, t = 1, \dots, T$$
(6)

where  $\mu_t$  is the trend,  $\psi$  is the cycle,  $\gamma_t$  is the seasonal, and  $\varepsilon_t$  is white noise.

Here, as in [65], STS models are automatically implemented in R by maximum likelihood through the function "StructTS" in the "stats" package. Further information on the automatically estimation and forecasting with STS models in R can be retrieved from [55].

#### 2.2.3. Neural Network Autoregression Model (NNAR)

Artificial neural networks (ANNs) have been shown to be able to mimic complicated real-world systems whilst effectively allowing for nonlinearities [66] (Pasini, 2015). The nodes, the network architecture reflecting the connections between nodes, and the training algorithm used to determine values of the network parameters for executing a certain task are the basic elements that describe an ANN [67]. Furthermore, an ANN is known as a multi-layered feed-forward network or multilayer perceptron (MLP) when each layer of nodes receives input from the preceding layer.

The models are called feed-forward because there are no feedback connections, and hence the model's forecasts (i.e., outputs) are not subsequently fed back into itself [68]. When working with time series data, lagged values of the time series are frequently employed as inputs to an ANN structure, resulting in neural network autoregression (NNAR) [69]. NNAR models for seasonal data are written as NNAR (p,P,k)m, where m is the seasonal period, p are nonseasonal lagged inputs for the linear AR process, P are seasonal lags for the AR process, and k is the number of hidden layer nodes.

The equation form of the NNAR model is given by:

$$Y = f(H) = f(W * X + B), X = [y(t-1), y(t-2), \dots, y(t-p)]$$
(7)

where Y is the output vector, X is the vector of inputs comprising the observed data's lagged values, H is the vector of nodes in the hidden layer, f is the activation function applied at H that is a transformation of a linear combination of the X (such as a sigmoid/logistic function), W is the weight matrix between X and H, and B is a bias vector [45].

The R forecast package's "nnetar" function, which is a network training function that changes weights and bias values during training, is used to train the feed-forward neural network. Through this algorithm, the nine univariate GHG emissions series are forecast by feed-forward neural networks (FFNN) with a single hidden layer and lagged inputs using the logit activation function to map the input value into the hidden node's value [70], as depicted in Figure 7.



Figure 7. A general form of a three-layer FFNN structure with lagged inputs.

Thus, the NNAR model uses *p*-lagged values of the GHG time series as inputs to a neural network with *k*-hidden nodes, for forecasting the output y(t) that contains a single node representing the predicted GHG emissions value. The function automatically makes 25 repetitions (i.e., trains 25 networks with random starting values) and selects the NNAR parameters (i.e., *p* and *P*) through minimizing the AIC, whereas the number of hidden notes is set as k = (p + P + 1)/2 (rounded to the nearest integer).

## 2.2.4. Forecasting Accuracy Metrics

The forecasting accuracy of the alternative predictive models for each of the nine GHG emissions series is assessed by estimating the Root Mean Squared Error (RMSE). RMSE is a scale-dependent Goodness-of-Fit (GoF) metric that, due to its benefits, emerged as one of the most popular [71]. Most importantly, RMSE carries the valuable advantage of direct interpretability in terms of measurement units. Although RMSE is less suitable when analyzing multiple time series of different measurement units [72], this is not the case in this research.

Thus, given its suitability and benefits relative to other GoF metrics, RMSE is the forecasting accuracy metric employed in the empirical investigation conducted in this study (similar approaches are found in [37,41,43]. RMSE equals the square root of the mean square error (MSE), and its estimation requires taking the differences between each point forecast and its corresponding value observed over the forecasting horizon, which is further squared and averaged as in Equation (8):

$$RMSE = \sqrt{\frac{1}{N}} \sum_{i=1}^{N} (y_i - \hat{y}_i)^2$$
(8)

2.2.5. Robustness Checks: The KSPA Predictive Accuracy Test

We scientifically reason that relying on the RMSE alone for identifying the best performing predictive model does not suffice to assure statistical robustness. We thus go one step further and check the statistical significance of point forecasts produced by the overperforming forecasting model  $M_i$ , i = 1, ..., 9 by estimating the Kolmogorov–Smirnov (KS) Predictive Accuracy test (KSPA). The KSPA test was developed by [73] as a complement statistical test to the Diebold–Mariano (DM) test [74,75] in order to verify the existence of significant differences between forecasts produced by two competing models.

Thus, whereas the DM approach [76] relies on the mean difference in errors, the KSPA approach is to evaluate the differences in the distribution of forecasting errors. The KSPA test brings some important benefits relative to the popular the Diebold–Mariano (DM) test, including more power, less sensitivity to outliers and increased understanding of the underlying distributional characteristics [73]. The two-sided KSPA test is automatically estimated through the "ks.test" function within R's "stats" package.

#### 3. Results

Table 2 reports the RMSE for out-of-sample forecasting results for the test period, or 2011–2018, thus, corresponding to an out-of-sample horizon of h = 8 steps ahead. Within the universe of seven competing models, the NNAR outperforms, showing a superior predictive ability 55% of the time, followed by the HW model (with a 22.22% score). Two statistical models (i.e., ETS and STS) were each found to over-perform in one out of nine instances. The other estimated models (i.e., ARIMA, TBATS, and the Naive model) were not able to produce accurate forecasts for GHG emissions in any of the nine CEE countries considered in this study.

	NNAR	ETS	ARIMA	STS	H-W	TBATS	Naive
Austria	2370.88	3673.57	4139.54	4442.13	4391.62	3492.11	4139.54
Bulgaria	4604.16	3847.74	4695.38	4207.66	2824.45	4492.23	3278.24
Croatia	2244.37	3160.71	4794.75	3167.93	3037.49	3105.58	3120.77
Czech Republic	8632.42	10,371.08	11,512.86	4985.65	4702.87	11,371.10	10,371.37
Hungary	3268.13	4345.55	9863.11	9168.60	9051.96	7938.42	4210.96
Poland	11,406.20	17,777.23	23,488.53	19,588.15	22,488.72	28,150.18	16,475.80
Romania	8348.96	4934.09	7904.91	18,473.16	18,262.47	8938.72	6668.08
Slovak Republic	3588.75	4003.57	4904.90	2301.25	2408.13	4776.77	4003.72
Slovenia	1856.16	1887.70	1887.73	2750.34	2623.70	1934.43	1888.18
Score *	5	1	0	1	2	0	0
Score (%) **	55.55%	11.11%	0%	11.11%	22.22%	0%	0%
Rank	1	3–4	5–7	3–4	2	5–7	5–7

Table 2. Prediction accuracy (RMSE for out-of-sample forecasting, forecasting horizon h = 8).

\* reflects the number of times (out of nine) that the model surpasses the other candidate models; \*\* is the percentage of outperformance (within nine iterations, or countries).

Table 3 reports the relative root mean squared error (RRMSE) results for the out-ofsample forecasts corresponding to the nine GHGs time series, where the NNAR model (i.e., the best-performing forecasting model over the lead-time) acts as a benchmark. Hence, the overall performance of NNAR across the nine series was proven to be 29% superior to the ARIMA forecast, 26% superior to TBATS, 15% superior to STS, 10% superior to the naive model, 8% superior to HW, and 7% superior to the ETS model when predicting the trend of GHG emissions in the nine CEE countries.

	NNAR/ETS	NNAR/ARIMA	NNAR/STS	NNAR/H-W	NNAR/TBATS	NNAR/Naive
Austria	0.65	0.57	0.53	0.54	0.68	0.57
Bulgaria	1.20	0.98	1.09	1.63	1.02	1.40
Croatia	0.71	0.47	0.71	0.74	0.72	0.72
Czech Republic	0.83	0.75	1.73	1.84	0.76	0.83
Hungary	0.75	0.33	0.36	0.36	0.41	0.78
Poland	0.64	0.49	0.58	0.51	0.41	0.69
Romania	1.69	1.06	0.45	0.46	0.93	1.25
Slovak Republic	0.90	0.73	1.56	1.49	0.75	0.90
Slovenia	0.98	0.98	0.67	0.71	0.96	0.98
Average	0.93	0.71	0.85	0.92	0.74	0.90
Score *	7	8	6	6	8	7

Table 3. The relative root mean squared error (RRMSE) for the forecasting horizon h = 8.

\* indicates the number of times (out of a maximum of nine) that NNAR outperforms the alternative predictive models.

#### Robustness Check: Predictive Accuracy Test

The Kolmogorov–Smirnov (KS) Predictive Accuracy test (KSPA) proposed by [73] additionally challenges NNAR's over-performance and contributes to ensuring the results' reliability. The KSPA test results are presented in Table 4 for each pair of competing models and each country, with NNAR as acting as a reference. In each instance where NNAR emerged as the top-performing model, the test identified significant disparities between the predictions issued by NNAR and the second-best performing model.

Table 4. The results of the Kolmogorov–Smirnov Predictive Accuracy (KSPA) test (p-values).

Country	KSPA ( <i>p</i> -Value)
Austria	0.01865 ***
Bulgaria	0.1871
Croatia	0.01865 **
Czech Republic	0.002486 *
Hungary	0.07937 ***
Poland	0.00235 **
Romania	0.002486 *
Slovak Republic	0.9801
Slovenia	0.08702 ***

Note: \* Based on the two-sided KSPA test at a 1% significance level, there is a statistically significant difference between the distribution of forecast errors from the best and second best performing models. \*\* indicates statistical significance at 5%. \*\*\* indicates statistical significance at 10%.

When NNAR did not outperform the best predictive model, the KSPA test was used to determine the differences between NNAR forecasts and the best predictive model. We can reject the null hypothesis and accept the alternate hypothesis when the two-sided KSPA test statistic is significant at 1%, demonstrating that the forecast errors from NNAR and the alternative model do not have the same distribution.

The KSPA test demonstrates that the NNAR forecasting technique outperforms its opponent for all countries (i.e., Austria, Croatia, Hungary, Poland, and Slovenia). The test, on the other hand, does not generally indicate the superiority of the rival model when NNAR is not shown to be the best model in terms of forecasting accuracy, with the exception of the Czech Republic and Romania. In the last stage of the current research, the over-performing predictive model for each GHG emissions series is fitted to the entire dataset N*i*, *i* = 1, ..., 9 and used to generate point estimates for GHGs in the nine countries, underlining the predicted levels for two relevant forecasting horizons, 2030 and 2050, respectively. As such, the entire forecasting horizon is set to h = 32, thus, reaching the 2050 benchmark. The in-sample fit is first confirmed by calculating the Ljung–Box test, which indicates that all models were properly specified (i.e., the residuals were checked for any signs of non-zero correlations at lags 1–20).

Table 5 reflects the percentage change relative to the 1990 level, as specified by the European Green Deal. The results indicate that, on average, the current decreasing trend in total GHGs in eastern European countries will continue over the next decades with a 26.48% decrease until 2030 and an overall 30.02% decrease across the nine countries until 2050. However, the estimation results indicate that two CEE countries (i.e., Austria and Slovenia) will register increases in total GHG emissions of 1.05% and 5.15%, respectively, by 2030 relative to their 1990 levels and similar levels of growth until 2050. Only a couple of countries are expected to reduce harmful emissions significantly and to meet the set pledges for 2030, i.e., Bulgaria (-56.11%) and Romania (-54.80%), although the GHG levels remain too high at the 2050 horizon.

Country	Point Forecast 2030	% (1990–2030)	Point Forecast 2050	% (1990–2050)
Austria	77,063.01	1.05	77,069.90	1.06
Bulgaria	43,162.67	-56.11	26,217.12	-73.34
Croatia	26,813.51	-6.54	25,091.39	-12.54
Czech Republic	110,686.81	-39.30	90,431.48	-50.41
Hungary	58,278.45	-34.61	58,273.77	-34.61
Poland	379,509.0	-14.20	379,508.6	-14.20
Romania	110,835.6	-54.80	110,968.8	-54.75
Slovak Republic	40,879.49	-38.98	42,461.97	-36.61
Slovenia	18,674.94	5.15	18,693.73	5.26
Average % change of total GHG emissions		-26.48%		-30.02%

Table 5. Forecasts for GHG emissions in nine CEE countries over 2019–2030 and 2019–2050.

Furthermore, we proceed to explore the preliminary performance of the over-performing models by assessing their forecasting accuracy for the two years subsequent to the dataset spanning period for which estimations of realized GHGs can be realized (i.e., 2019 and 2020). Thus, although official statistics for total GHG emissions are not yet available, data on  $CO_2$  emissions for 2019 and 2020 are available for most of the countries included in our data sample.

Thus, in order to provide an estimation for the realized GHG levels in 2019 and 2020, the following approach was implemented: first, the growth rate in  $CO_2$  emissions for the years 2019 and 2020 was extracted from  $CO_2$  data, and then the same growth rate was applied to the GHG 2018 level. Thus, estimations of total GHGs for 2019 and 2020 were produced by assuming that the total GHGs will register a similar evolution to  $CO_2$  emissions. Albeit imperfect, we scientifically reason that this approach is able to provide a sufficiently accurate approximation of the realized GHGs values and can capture the trend particularly well. In a few instances where national reports or other national sources that provide information for actual GHG data were found, the data extracted from those sources were employed.

Table 6 summarizes these findings and indicates all specific sources. The results show high accuracy of the forecasting models in predicting GHG emissions values for the year 2019, albeit, as expected, they overestimated GHG values for the year 2020, when the pandemic was a black-swan event that significantly affected the global GHG levels in 2020. As a result, the confidence level decreased from 98.5% for 2019 to 90% for the year 2020, registering an average value of approximately 95% over the two years. Considering the unseen effects of the COVID-19 pandemic, it can be extracted that the over-performing models are able to provide accurate forecasts for emission trends during "normal" conditions and thus can assist the policymaking process at the EU level.

	Actual Value (A)	Forecasted Value (F)	Error (A–F)	Error /A
Austria 2019	79,800 <sup>1</sup>	77,845.6	1954.4	0.024
Austria 2020	73,600 <sup>1</sup>	76,331.2	-2731.2	-0.037
Bulgaria	57,200 <sup>2</sup>	55,829.01	1371	0.024
	50,724 <sup>3,*</sup>	57,318	-6594	-0.13
Croatia	22,652 <sup>3,*</sup>	23,172	-520	-0.023
	21,633 <sup>3,*</sup>	24,039	-2406	-0.111
Czech Republic	107,194.07 <sup>3</sup> ,*	123,123	-15,928.9	-0.149
	101,036.45 3,*	125,088	-24,051.5	-0.238
Hungary	60,800 <sup>4</sup> ,*	60,344	456	0.008
	58,368 <sup>4</sup> ,*	59,402	-1034	-0.018
Poland	394,000 <sup>4</sup> ,*	384,761	9239	0.023
	373,680 <sup>5</sup>	381,685	-8005	-0.021
Romania	105,997.16 <sup>3</sup> ,*	109,402	-3404.84	-0.032
	101,888 <sup>3</sup> ,*	109,715	-7827	-0.077
Slovak Republic	41,900 <sup>6</sup>	40,909	991	0.024
	35,537.7 <sup>4</sup> ,*	39,716	-4178.3	-0.118
Slovenia	16,601 <sup>3</sup> ,*	17,145	-544	-0.033
	14,883 <sup>3,*</sup>	17,191	-2308	-0.155
MAPE (total)				0.058 (94.2% confidence level)
MAPE 2019				0.015 (98.5% confidence level)
MAPE 2020				0.10 (90.0% confidence level)

**Table 6.** Preliminary assessment of the forecasting performance for aggregated GHGs at the national level (2019 and 2020).

\* Estimated value from CO<sub>2</sub> emissions trends. <sup>1</sup> Source: Umweltbundesamt. <sup>2</sup> Source: https://www.europarl. europa.eu/RegData/etudes/BRIE/2021/689330/EPRS\_BRI(2021)689330\_EN.pdf (accessed on 11 February 2022). <sup>3</sup> Source: Our World in Data. <sup>4</sup> Source: Statista. <sup>5</sup> Source: http://seo.org.pl/en/w-2020-r-emisja-gazowcieplarnianych-w-polsce-spadla-do-37368-tys-kt-ekw-co2/ (accessed on 11 February 2022). <sup>6</sup> Source: https: //www.europarl.europa.eu/RegData/etudes/BRIE/2021/698767/EPRS\_BRI(2021)698767\_EN.pdf (accessed on 11 February 2022).

Figure 8 reflects the GHG forecasts over the 2019–2050 period and also highlights the EU Green Deal 2030 individual targets in terms of the total GHG emissions levels to be achieved by 2030. The countries that are farthest from meeting the targets (in absolute terms) are Austria and Poland, whereas the Czech Republic will continue on the path toward pollution mitigation; albeit, the set targets are not projected to be met.



Figure 8. Cont.





Slovenia Total GHG emissions forecasts (2019–2050)





On the other hand, the Slovak Republic is set to reverse its progress toward emissions mitigation, whereas Austria, Croatia, Hungary, Poland, and Slovenia show a future constant evolution in the total GHG emissions. Romania and Bulgaria are the only countries in the CEE area that are expected to meet their 2030 emission mitigation targets.

### 4. Discussion

Our research findings confirm that the neural network autoregressive model (NNAR) exhibited the best out-of-sample predictive ability within the universe of statistical and deep-learning models employed in this study, being able to achieve the lowest forecast error for the nine GHG emissions series 55% of the time. This is in line with [77], who concluded on the efficiency of artificial neural networks (ANN) in time series modeling when a variable's previous values were employed as inputs to describe the future values. Our results also support the conclusion of [43–45], which also identify NNAR as overperforming in forecasting polluting (CO<sub>2</sub> and/or GHG) emissions.

Next, we report the total GHG emissions values forecasted with the best predictive model identified for each country. Overall, the estimation results indicate that future emission forecast levels are insufficient to meet the 2050 targets at the CEE level (i.e., carbon neutrality). Our findings thus back the findings of [22] and indicate that the EU climate goals are overly optimistic. A preliminary assessment of the forecasting ability over the

first two time indices in the forecasting horizon (i.e., the years 2019–2020) indicates the high accuracy of the forecasting models in predicting emission values for the year 2019; albeit, as expected, they could not predict the black-swan event that significantly affected the global GHG levels in 2020.

However, it should be mentioned that, to reach carbon neutrality, countries can use carbon credits or offsets from projects that reduce, avoid, or temporarily capture GHGs [78]. Further, net-zero emissions are achieved when all produced GHG emissions are offset by removing GHGs from the atmosphere in a process known as carbon removal. Thus, although carbon neutrality does not imply absolute net-zero emissions, it does, however, require the mitigation of GHGs to as close to zero as possible [79]. Moreover, the most cost-efficient way to address GHG concentrations is by avoiding emissions [80].

One of the main reasons for the previous reduction in GHG emissions within the EU is the increased reliance on renewable energy sources. However, the technical complexities of integrating green sources increase with their share in power generation, and transport, heating, industry, and agriculture are more difficult to decarbonize than the electricity sector, which has been mainly accomplished thus far [81]. Consequently, this might explain why the CEE countries analyzed in this study are projected to miss the ambitious emission targets set through the European Green Deal.

This further confirms that the overwhelming dependency on fossil fuels of CEE economies remains a considerable roadblock to the region's decarbonization goals [82]. In addition, the findings indicate that the 2030 EU-set 55% emissions reduction targets relative to 1990 levels will only be met by Bulgaria and Romania, whereas other countries are expected to continue the decreasing trend without meeting the goal, or even see increasing emissions levels over the next decade.

Our results are thus in line with [83] and confirm that Romania and Bulgaria are committed to mitigating pollution and pursuing low-carbon development [84]. Furthermore, it should be noted that Romania has already met its 2020 renewables target of 24 percent of the final energy consumption from renewables well before the deadline and is also on the right track to reach its 2030 renewables target of 30.7% [85], which helps to explain both the past decreasing GHG emissions trend and its projected evolution.

A similar situation is encountered in Bulgaria, as the country has reached its renewable energy legally binding target for 2020 (16%) since 2013 [86]. Additionally, it was demonstrated that a power system with a substantially higher deployment of renewable energy sources is feasible [87], indicating that more impactful measures can be implemented to accelerate the emissions mitigation trend.

As such, we agree with [88] that an important economic goal is to expedite the transition to renewable energy. This is also in line with previous studies (among others, [89–94] that reported that renewable energy and polluting emissions have a negative association.

Moreover, it should also be mentioned that both Romania and Bulgaria have suffered significant decreases in their population number over the past decades, mostly due to low birth rates and a chronic migration problem (i.e., Romania has the largest negative net migration stocks with the rest of the EU [95]). The demographic decline is also expected to continue until 2050, further explaining the GHG emissions mitigation projected for the countries [96].

However, it should also be acknowledged that, unlike the majority of their former Soviet bloc counterparts, Bulgaria and Romania have not been able to successfully reap the benefits of their industries collapsing with the demise of communism after 1989, which supplied them with millions of tons of excess rights known as Assigned Amount Units (AAUs) [97]. This, in turn, might also explain their more intensive reduction in total GHGs after 1990 relative to other CEE countries.

Nonetheless, the required reduction in greenhouse gas emissions that CEE countries must accomplish comes with important economic costs, which are non-trivial for the more vulnerable EU members [98,99]. Particularly, Poland and the Czech Republic—which are, with Germany, Europe's top three coal-burning countries [100]—will face significant

challenges. Although it can be argued that health benefits can outweigh the costs [101], we take a similar view with [102] and suggest that financial aid and technology transfers are required to support efforts toward carbon neutrality.

The European Union economic recovery package addressing the COVID-19 pandemic's economic and social consequences, i.e., the Next Generation EU (NGEU) fund ([103–105]) should be directed toward implementing targeted climate change and green transition measures and be aimed at achieving a synchronized EU recovery.

To this end, effective measures can emerge from further research based on the current findings and using geo-spatial GHG data for countries that are projected to significantly miss policy targets under the status quo hypothesis. This supports the claims of [106], showing that it is frequently required to collect data not only on the level at which a policy is implemented but also below this level. This would further permit addressing the heterogeneous nature of EU regions and their distinct characteristics and sensitivities to specific policies.

Finally, this research fully agrees with [107] and highlights that increasing ongoing efforts to integrate statistics and geo-spatial data and thus to provide disaggregation at the highest possible level of detail would be conducive to more integrated research endeavors capable of providing reliable and relevant information for effective and efficient decision making, and ultimately support the worldwide goal of sustainable development.

For an exemplification of the aforementioned conclusion, we take a closer look at the country that is most expected to miss the EU-set targets, i.e., Poland. Coincidentally, Poland constitutes a rare case where governmental geospatial data (i.e., a relevant network of air pollution sensors) is publicly available, thus, providing a unique opportunity to identify pollution-related vulnerable areas. Consequently, we sourced pollution data from three government stations located in the three most important Polish urban areas, i.e., Warsaw, Krakow, and Lodz, respectively.

The data includes hourly observations and spans 8 February 2022 to 9 March 2022, for a total of 720 records (i.e.,  $24 \times 30$ ) for each of the following air pollutants: PM10, PM2.5, NO<sub>2</sub>, C<sub>6</sub>H<sub>6</sub>, and CO. The dataset employed for the geo-spatial analysis thus comprises a total of 3600 hourly records. Appendix A Figure A1a–c zooms in at the city level to highlight the exact geo-location of the governmental sensor for which air quality measurement data were extracted. Table 7 centralizes the statistics for all measured parameters by the official government sensors located in the three reference stations over the most recent month.

The statistics in Table 7 demonstrate that the distribution of pollutant data in the three polish urban areas from February to March 2022 was heterogenic, highlighting that Poland's main air quality problem is located in Krakow. The current EU standards that members should meet are, respectively, 40  $\mu$ g/m<sup>3</sup> (1-year averaging period) for PM10 concentrations, 20  $\mu$ g/m<sup>3</sup> (1-year averaging period) for PM2.5 concentrations, 40  $\mu$ g/m<sup>3</sup> (1-year averaging period) for NO<sub>2</sub> concentrations, 10 mg/m<sup>3</sup> (maximum daily 8-hour mean) for CO, and 5  $\mu$ g/m<sup>3</sup> (1-year averaging period) for benzene (C<sub>6</sub>H<sub>6</sub>) [109].

As a result, Krakow is the only area in Poland (among the three sampled here) that currently fails to meet the EU air quality standards for PM10, PM2.5, and NO<sub>2</sub>, with adverse health effects for its population. This supports [110] and is also in line with a 2016 WHO report (cited by [111]) that positions Krakow as the eighth worst city in the European Union (EU) when it comes to surpassing the thresholds for the air concentration of particulate matter. Figure 9, which offers an overview of pollutant concentration at the country level, further confirms that Poland's pollution is mostly concentrated in and around the city of Krakow.

We next attempt to forecast one of the most dangerous pollutants, i.e., particulate matter 2.5 (PM2.5) for the selected governmental air sensor in Krakow, and we investigate how far the air quality in Krakow will remain from the EU standards over the short term. Particulate matter (PM) refers to suspensions in the air comprising coarse, fine, and ultrafine solid and liquid particles, whereas PM2.5 refers to PM with a diameter of less than  $2.5 \,\mu\text{g/m}^3$  (i.e., standard air quality measurements describe PM concentrations in terms

of micrograms per cubic meter or  $\mu g/m^3$ ) that remain suspended in the air for prolonged periods and can cause a variety of short- and long-term health problems. Long-term exposure to PM2.5 has been found to cause, among others, chronic respiratory issues, heart disease, and cancers [39,113]. Overall, it is estimated that particulate matter smaller than 2.5 micrometers (PM2.5) ambient air pollution caused approximately 4.2 million deaths in 2015 ([114,115].

**Table 7.** Geo-spatial data analysis from selected Polish government pollution measurements(8 February 2022–9 March 2022).

Components	PM10	PM2.5	NO <sub>2</sub>	$C_6H_6$	СО
Unit	μg/m <sup>3</sup>	μg/m <sup>3</sup>	μg/m <sup>3</sup>	μg/m <sup>3</sup>	mg/m <sup>3</sup>
Station: Łodz (ul. Gdańska 16)					
Min	0.5	NA	2.2	0.3	0.2
Max	136.5	NA	82.1	10	2.4
Mean	20.7	NA	21.8	1.6	0.5
Station: Warsaw (al. Niepodległości)					
Min	5.9	2.3	5.4	0.3	0.2
Max	183.2	87.4	101.3	4	1.3
Mean	35.1	16.9	42.7	1.2	0.5
Station: Krakow (al. Krasińskiego)					
Min	3.4	2.2	6.2	0.1	0.2
Max	215.4	117.2	119.8	7.9	2
Mean	52.4	27.6	51.4	1.4	0.7

Source of data: Poland's Chief Inspectorate for Environmental Protection database (http://powietrze.gios.gov.pl/pjp/home [108] (accessed on 9 March 2022)).



**Figure 9.** Poland air quality concentration map on 9 March 2022. Source: Snapshot of the Map of Air Quality by Airly: https://map.airly.org/ [112] (accessed on 10 March 2022)). Green color: good air quality (low level of pollutants). Yellow color: medium air quality (medium level of pollutants. Orange color: low air quality (high level of pollutants). Red color: very low air quality (very high level of pollutants).

Consequently, we collected hourly PM2.5 levels measured by the government set-up monitoring station over the previous year, thus, constructing a dataset of 8782 hourly observations over the 8 March 2021 (02:00)–8 March 2022 (03:00) period. Figure 10 shows the distribution of the pollutant during this timeframe, reflecting that it remains above the EU-set standards for most of the year, with peaks during the winter months, which in turn indicates that the burning of low-quality coal in coal-fired stoves is an important cause of environmental pollution in the city, in line with [116].





Created with Datawrapper

**Figure 10.** The trend of PM2.5 in Krakow during 8 March 2021 (02:00)–8 March 2022 (03:00). Source of data: Poland's Chief Inspectorate For Environmental Protection (http://powietrze.gios.gov.pl/pjp/home (accessed on 9 March 2022)). Authors' representation in Datawrapper.

For estimation purposes, the hourly dataset is subsequently divided into a training set containing the first 7000 observations (i.e., approximately 80% of the data) and a test set of 1782 observations (i.e., 20% of the data) that is employed for assessing the out-of-sample forecasting accuracy of the seven alternative automated forecasting models. Table 8 reports the accuracy metrics for the out-of-sample forecasting of PM2.5, indicating that NNAR is again capable of producing the most accurate out-of-sample forecasts.

**Table 8.** Prediction accuracy metrics (ME and RMSE) for the out-of-sample forecasting \* of PM2.5 in Krakow.

Predictive Model	ME	RMSE
NNAR	5.09	26.93
TBATS	13.00	28.26
ARIMA	18.92	31.42
Naive	20.59	32.44
STS	22.08	33.40
EXP	-54.12	73.95
HW	94.70	106.59

\* Forecasting horizon of 1782 h (or approximately 74 days); training period of 7000 h (approximately 292 days).

Lastly, the best performing forecasting algorithm (i.e., NNAR) is fitted to the entire dataset containing 8782 hourly measurements for PM2.5 in the Krakow area and instructed to produce forecasted values for PM2.5 over the next 30 days (or 720 hourly observations). Figure 11 reflects the estimation results, along with the benchmark 1-year average level of  $20 \ \mu g/m^3$  established at the EU level that should be met by all EU members, including Poland.



**Figure 11.** Short-term predictions of PM2.5 in Krakow issued by NNAR(26,14). Source: estimation results. Model Information: Average of 20 networks, each of which is a 26-14-1 network with 393 weight options.

The research results indicate without a doubt that, absent of further policy interventions, the air quality in Krakow will remain low with subsequent negative public health consequences. Thus, forecasts show an upward trend that is a cause for alarm, given that the current quality of PM2.5 levels is already significantly higher than the safe limit of  $20 \ \mu g/m^3$  prescribed by the EU, which is, in turn, significantly higher than the PM2.5 limit set by the WHO, which recommends a maximum annual mean of  $5 \ \mu g/m^3$  [113].

Consequently, our research highlights that Polish authorities should take targeted actions to diminish pollution in Krakow. In this respect, we support the conclusion of [111] and agree that, among other actions, informative campaigns and education could make a significant impact on pollution mitigation at the city level. In line with [116], we reason that other measures could focus on restrictions on the use of solid fuel systems, expanding and modernizing heating networks, eliminating waste incineration, and limiting emissions from transportation.

The current research findings further show that failure to implement efficient pollutionreduction actions would result in Poland missing the 2030 and 2050 EU targets with significant negative consequences. Similarly, the availability of granular data for other countries that are also expected to struggle to meet the mandatory pollution-mitigation targets given the status quo (i.e., especially Austria and Slovenia) would prove of great value, as a similar geospatial analysis would pinpoint specific geo-locations where targeted interventions are needed and thus would significantly mitigate the risk of incurring noncompliance-related sanctions.

#### 5. Conclusions

The European Union is dedicated to leading the worldwide battle against climate change. Already a lesser contributor to world pollution than the world average, the EU recognized the need to go even further with its climate ambitions to transform Europe into a highly energy-efficient, low-carbon economy. In December 2020, EU leaders agreed to a binding EU goal of a net domestic reduction in greenhouse gas emissions of at least 55 percent by 2030 compared to 1990 and to reach carbon neutrality by 2050. However, reaching these ambitious goals poses significant challenges for the more vulnerable, fossilfuel-dependent CEE economies, whereas the failure to reach the set pledges might trigger infringement procedures, legal challenges, and ultimately financial penalties.

In this context, accurate predictions of GHG emissions in CEE countries are crucial, which motivated this study. Estimations show that, by 2030, the nine CEE countries would register, on average, a 20% reduction in GHG emissions compared with the 1990 base-year emissions, albeit these reductions remain insufficient compared to the 2030 55% mitigation target. The only two CEE countries that are expected to meet their emission mitigation goals for 2030 are Romania and Bulgaria, while none of the CEE countries included in the analysis are expected to reach a net-zero target by 2050 by mitigating total GHG emissions to close to zero values in absolute terms. Austria and Slovenia are the farthest from attaining the 2030 objective, while Poland (in absolute terms) and Slovenia (in relative terms) are the least likely to achieve the climate policy aims set for 2050.

Hence, our research findings indicate that achieving carbon neutrality will only be possible through heavy reliance on carbon credits. However, it should be mentioned that forecasting models run under the status quo hypothesis, and thus any future developments are not considered when producing estimates. Consequently, we acknowledge that, given the long forecasting horizon employed in this study, all estimated models should be reviewed and updated regularly to ensure that they remain appropriate for the dynamic circumstances.

Particularly, updating the time series with the inclusion of observations corresponding to the COVID-19 pandemic should lead to more accurate estimations of GHG trends. This study brings evidence based on existing conditions that the current pollution mitigation measures implemented in the selected countries are insufficient, and reaching 2030 targets will require additional efforts that go beyond the status quo.

Nonetheless, a successful EU transition toward carbon neutrality should consider the diverse power systems and starting points in the decarbonization process across CEE member states and should be complemented by adequate social protection programs. Furthermore, as the transition of Central and Eastern European countries to mature and socially just market economies is still underway, the EU's post-pandemic recovery plan should be efficiently used to successfully achieve this transition.

In this respect, this study serves as the foundation for future research aimed at uncovering specific areas that require intervention through research on finer scale geospatial GHG data, particularly for countries projected to significantly miss the EU net-zero policy targets. Currently, this study accomplished monitoring and evaluation of the EU low-carbon policy implementation and identified the requirements for policy response at a national level.

We also performed geospatial analysis for Poland and found that the distribution of the main air pollutants within three major urban areas was highly heterogenic, as a significantly high concentration of pollution was encountered in and around the city of Krakow. Projections of PM2.5 in the city of Krakow showed that the air quality in Krakow will remain too low relative to EU and WHO standards and that immediate pollutionreduction measures are paramount to protect population health and avoid the negative consequences of failing to meet the 2030 and 2050 EU binding targets. For other at-risk CEE countries according to the current findings, further research is needed to fill in the missing pieces once specific geo-spatial data becomes available.

Consequently, we reason that statistical and geo-spatial views complement one another and that additional efforts should be directed toward accelerating their integration. Similarly, research on aggregated statistical data and on (geo) spatial data complement and enhance each other. Thus, the integration of data would, in turn, accommodate integrated research capable of identifying specific issues that require policy responses and thus better assist an effective and efficient decision-making process.

Ultimately, the integrated framework can support the worldwide goal of sustainable development.

Author Contributions: Conceptualization, Cristiana Tudor; methodology, Cristiana Tudor and Robert Sova; software, Cristiana Tudor and Robert Sova; validation, Cristiana Tudor; formal analysis, Cristiana Tudor; investigation, Cristiana Tudor; data, Cristiana Tudor; writing—original draft preparation, Cristiana Tudor; writing— review and editing, Cristiana Tudor and Robert Sova; visualization, Cristiana Tudor; supervision, Cristiana Tudor and Robert Sova; project administration, Cristiana Tudor All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

25 of 30

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** GHG emissions data is available from the World Bank's Development Indicators (WDI) database. Data for air pollutants measured by local sensors in Poland are available from the Poland's Chief Inspectorate For Environmental Protection database.

Conflicts of Interest: The authors declare no conflict of interest.

## Appendix A



(a) Location of the air quality sensor in the Warsaw urban area



(b) Location of the air quality sensor in the Lodz urban area



(c) Location of the air quality sensor in the Krakow urban area

**Figure A1.** Location of the air quality sensors in Poland's three main urban areas: (a) Warsaw, (b) Lodz, and (c) Krakow. Source: Snapshot from Poland's Chief Inspectorate For Environmental Protection (http://powietrze.gios.gov.pl/pjp/home (accessed on 9 March 2022)). Green color: good air quality (low level of pollutants). Yellow color: medium air quality (medium level of pollutants. Orange color: low air quality (high level of pollutants). Red color: very low air quality (very high level of pollutants).

# References

- 1. European Commission (EC). A European Green Deal. 2021. Available online: https://ec.europa.eu/info/strategy/priorities-2019 -2024/european-green-deal\_en (accessed on 11 December 2021).
- European Commission (EC). Eurobarometer Survey: Europeans Consider Climate Change to Be the Most Serious Problem Facing the World. 2021. Available online: https://ec.europa.eu/commission/presscorner/detail/en/ip\_21\_3156 (accessed on 11 December 2021).
- 3. European Commission (EC). What Is the European Green Deal? 2019. Available online: https://ec.europa.eu/commission/ presscorner/detail/en/fs\_19\_6714 (accessed on 11 December 2021).
- 4. EU Monitor. Communication from the Commission to the European Parliament, the European Council, the Council, the European Economic and Social Committee and the Committee of the Regions, The European Green Deal, COM (2019) 640 Final. 2019. Available online: https://www.eumonitor.eu/9353000/1/j9vvik7m1c3gyxp/vl4cnhyp1ort (accessed on 11 December 2021).
- 5. Teevan, C.; Medinilla, A.; Sergejeff, K. The Green Deal in EU foreign and development policy. In *ECDPM Briefing Note* 131; ECDPM: Maastricht, The Netherlands, 2021.
- Climate Change News. European Green Deal Must Be Central to a Resilient Recovery after COVID-19. 2020. Available online: https://www.climatechangenews.com/2020/04/09/european-green-deal-must-central-resilient-recovery-covid-19/ (accessed on 11 December 2021).
- Sikora, A. European Green Deal-legal and financial challenges of the climate change. In *ERA Forum*; Springer: Berlin/Heidelberg, Germany, 2021; Volume 21, pp. 681–697.
- 8. Kougias, I.; Taylor, N.; Kakoulaki, G.; Jäger-Waldau, A. The role of photovoltaics for the European Green Deal and the recovery plan. *Renew. Sustain. Energy Rev.* 2021, *Vol* 144, 111017. [CrossRef]
- EU Commission. Stepping up Europe's 2030 Climate Ambition: Investing in a Climate-Neutral Future for the Benefit of Our People. 2020. Available online: https://knowledge4policy.ec.europa.eu/publication/communication-com2020562-steppingeurope\T1\textquoterights-2030-climate-ambition-investing-climate\_en (accessed on 11 December 2021).
- 10. World Economic (WE) Forum. What You Need to Know about the European Green Deal-and What Comes Next. 2021. Available online: https://www.weforum.org/agenda/2021/07/what-you-need-to-know-about-the-european-green-deal-and-what-comes-next/ (accessed on 5 December 2021).
- 11. European Commission. European Climate Law. 2021. Available online: https://ec.europa.eu/clima/eu-action/european-green-deal/european-climate-law\_en (accessed on 11 December 2021).
- 12. Economic Commission. The European Union Continues to Lead Global Fight Against Climate Change. 2019. Available online: https://ec.europa.eu/clima/news-your-voice/news/european-union-continues-lead-global-fight-against-climate-change-2019-09-11\_de (accessed on 5 December 2021).
- United Nations Framework Convention on Climate Change. The Paris Agreement. 2021. Available online: https://unfccc.int/ process-and-meetings/the-paris-agreement/the-paris-agreement (accessed on 14 December 2021).
- Crippa, M.; Guizzardi, D.; Solazzo, E.; Muntean, M.; Schaaf, E.; Monforti-Ferrario, F.; Banja, M.; Olivier, J.; Grassi, G.; Rossi, S.; et al. *GHG Emissions of All World Countries*; EUR 30831 EN; Publications Office of the European Union: Luxembourg, 2021; ISBN 978-92-76-41547-3. [CrossRef]
- United Nations Climate Change. Nationally Determined Contributions (NDCs). 2021. Available online: https://unfccc.int/ process-and-meetings/the-paris-agreement/nationally-determined-contributions-ndcs/nationally-determined-contributionsndcs (accessed on 12 December 2021).
- Sustainable Development Solutions Network (SDSN). European Commission Launches Proposals to Reach 55% Emissions Reduction by 2030. 2021. Available online: https://www.unsdsn.org/european-commission-launches-proposals-to-reach-55 -emissions-reduction-by-2030 (accessed on 11 December 2021).
- 17. Pavlínek, P.; Ženka, J. Upgrading in the automotive industry: Firm-level evidence from Central Europe. *J. Econ. Geogr.* **2010**, *11*, 559–586. [CrossRef]
- Demeter, K.; Szász, L. The diversity of European manufacturing plant roles in international manufacturing networks. J. East Eur. Manag. Stud. 2016, 21, 184–208. [CrossRef]
- 19. Pellényi, G.M. The Role of Central and Eastern Europe in Global Value Chains: Evidence from Occupation-Level Employment Data (Economic Brief No. 062); Directorate General Economic and Financial Affairs (DG ECFIN), European Commission: Brussels, Belgium, 2020.
- The Economist, Eastern Europe's COVID-19 Recession Could Match Its Post-Communist One. Available online: https://www. economist.com/europe/2020/05/28/eastern-europes-covid-19-recession-could-match-its-post-communist-one (accessed on 11 December 2021).
- McKinsey. COVID-19: An Inflection Point for Industry 4.0. 2021. Available online: https://www.mckinsey.com/businessfunctions/operations/our-insights/covid-19-an-inflection-point-for-industry-40 (accessed on 11 December 2021).
- 22. European Environment Agency (EEA). Total Greenhouse Gas Emission Trends and Projections in Europe. 2021. Available online: https://www.eea.europa.eu/ims/total-greenhouse-gas-emission-trends (accessed on 20 December 2021).
- 23. The International Comparative Legal Guides (ICGL). EU Green Deal Terms and Challenges. 2021. Available online: https://iclg.com/news/16111-eu-green-deal-terms-and-challenges (accessed on 11 February 2022).

- 24. POLITICO. Making the Green Deal Real. 2020. Available online: https://www.politico.eu/article/wonk-guide-from-europeangreen-deal-to-real/ (accessed on 11 December 2021).
- Energy Monitor. The Myths Keeping Eastern Europe in the Fossil Age. 2020. Available online: https://www.energymonitor.ai/ policy/the-myths-keeping-eastern-europe-in-the-fossil-age (accessed on 20 December 2021).
- Alves Dias, P.; Kanellopoulos, K.; Medarac, H.; Kapetaki, Z.; Miranda Barbosa, E.; Shortall, R.; Czako, V.; Telsnig, T.; Vazquez Hernandez, C.; Lacal Arantegui, R.; et al. *EU Coal Regions: Opportunities and Challenges Ahead*; EUR 29292 EN; Publications Office of the European Union: Luxembourg, 2018; ISBN 978-92-79-89884-6. [CrossRef]
- Stockholm School of Economics. Green Concerns and Salience of Environmental Issues in Eastern Europe. 2021. Available online: https://www.hhs.se/en/about-us/news/site-publications/publications/2021/green-concerns-and-salienceof-environmental-issues-in-eastern-europe/ (accessed on 20 December 2021).
- Lorenzoni, I.; Pidgeon, N.F. Public Views on Climate Change: European and USA Perspectives. *Clim. Chang.* 2006, 77, 73–95. [CrossRef]
- 29. Poortinga, W.; Whitmarsh, L.; Steg, L.; Böhm, G.; Fisher, S. Climate change perceptions and their individual-level determinants: A cross-European analysis. *Glob. Environ. Chang.* **2019**, *55*, 25–35. [CrossRef]
- 30. Marquart-Pyatt, S.T.; Qian, H.; Houser, M.K.; McCright, A.M. Climate Change Views, Energy Policy Preferences, and Intended Actions Across Welfare State Regimes: Evidence from the European Social Survey. *Int. J. Sociol.* **2019**, *49*, 1–26. [CrossRef]
- 31. Wen, L.; Yuan, X. Forecasting CO<sub>2</sub> emissions in Chinas commercial department, through BP neural net-work based on random forest and PSO. *Sci. Total Environ.* **2020**, *718*, 137194. [CrossRef] [PubMed]
- Mahajan, S.; Chen, L.J.; Tsai, T.C. An empirical study of PM2. 5 forecasting using neural network. In 2017 IEEE SmartWorld, Ubiquitous Intelligence & Computing, Advanced & Trusted Computed, Scalable Compu-Ting & Communications, Cloud & Big Data Computing, Internet of People and Smart City Innovation (Smart-World/SCALCOM/UIC/ATC/CBDCom/IOP/SCI); IEEE: Piscataway, NJ, USA, 2017; pp. 1–7.
- Mardani, A.; Streimikiene, D.; Cavallaro, F.; Loganathan, N.; Khoshnoudi, M. Carbon dioxide (CO<sub>2</sub>) emissions and economic growth: A systematic review of two decades of research from 1995 to 2017. *Sci. Total Environ.* 2019, 649, 31–49. [CrossRef] [PubMed]
- 34. Kampa, M.; Castanas, E. Human health effects of air pollution. Environ. Pollut. 2008, 151, 362–367. [CrossRef] [PubMed]
- 35. Anderson, H.R. Air pollution and mortality: A history. *Atmos. Environ.* **2009**, *43*, 142–152. [CrossRef]
- 36. Di, Q.; Wang, Y.; Zanobetti, A.; Wang, Y.; Koutrakis, P.; Choirat, C.; Dominici, F.; Schwartz, J.D. Air pollution and mortality in the Medicare population. *N. Engl. J. Med.* **2017**, *376*, 2513–2522. [CrossRef]
- 37. Hassani, H.; Silva, E.S.; Gupta, R.; Segnon, M.K. Forecasting the price of gold. Appl. Econ. 2015, 47, 4141–4152. [CrossRef]
- Breiman, L. Statistical modeling: The two cultures (with comments and a rejoinder by the author). *Stat.-Cal Sci.* 2001, *16*, 199–231.
   [CrossRef]
- Nath, P.; Saha, P.; Middya, A.I.; Roy, S. Long-term time-series pollution forecast using statistical and deep learning methods. *Neural Comput. Appl.* 2021, 33, 12551–12570. [CrossRef] [PubMed]
- Meng, M.; Niu, D. Modeling CO<sub>2</sub> emissions from fossil fuel combustion using the logistic equation. *Energy* 2011, *36*, 3355–3359. [CrossRef]
- Silva, E.S. A combination forecast for energy-related CO<sub>2</sub> emissions in the United States. *Int. J. Energy Stat.* 2013, *1*, 269–279. [CrossRef]
- Malik, A.; Hussain, E.; Baig, S.; Khokhar, M.F. Forecasting CO<sub>2</sub> emissions from energy consumption in Pakistan under different scenarios: The China–Pakistan economic corridor. *Greenh. Gases Sci. Technol.* 2020, 10, 380–389. [CrossRef]
- Tudor, C. Predicting the evolution of CO<sub>2</sub> emissions in Bahrain with automated forecasting methods. *Sustainability* 2016, *8*, 923. [CrossRef]
- 44. Qader, M.R.; Khan, S.; Kamal, M.; Usman, M.; Haseeb, M. Forecasting carbon emissions due to electricity power generation in Bahrain. *Environ. Sci. Pollut. Res.* 2022, 29, 17346–17357. [CrossRef] [PubMed]
- 45. Tudor, C.; Sova, R. Benchmarking GHG Emissions Forecasting Models for Global Climate Policy. *Electronics* **2021**, *10*, 3149. [CrossRef]
- 46. Bai, X.; Dawson, R.; Ürge-Vorsatz, D.; Delgado, G.C.; Barau, A.S.; Dhakal, S.; Dodman, D.; Leonardsen, L.; Masson-Delmotte, V.; Roberts, D.C.; et al. Six research priorities for cities and climate change. *Nature* **2018**, *555*, 23–25. [CrossRef] [PubMed]
- 47. Boychuk, P.; Boychuk, K.; Nahorski, Z.; Horabik, J. Spatial inventory of greenhouse gas emissions from the road transport in Poland. *ECONTECHMOD Int. Q. J. Econ. Technol. Model. Processes* **2012**, *1*, 9–15.
- 48. Oda, T.; Maksyutov, S. A very high-resolution (1 km × 1 km) global fossil fuel CO<sub>2</sub> emission inventory derived using a point source database and satellite observations of nighttime lights. *Atmos. Chem. Phys.* **2011**, *11*, 543–556. [CrossRef]
- 49. Danylo, O.; Bun, R.; See, L.; Charkovska, N. High-resolution spatial distribution of greenhouse gas emissions in the residential sector. *Mitig. Adapt. Strateg. Glob. Chang.* **2019**, *24*, 941–967. [CrossRef]
- 50. Tashman, L.J. Out-of-sample tests of forecasting accuracy: An analysis and review. Int. J. Forecast. 2000, 16, 437–450. [CrossRef]
- 51. Schneider, J. Cross Validation. 1197. Available online: https://www.cs.cmu.edu/~{}schneide/tut5/node42.html (accessed on 20 December 2021).
- R Core Team. R: A Language and Environment for Statistical Computing; R Foundation for Statistical Computing: Vienna, Austria, 2013; ISBN 3-900051-07-0. Available online: http://www.R-project.org/ (accessed on 5 October 2021).

- 53. Hyndman, R.J.; Khandakar, Y. Automatic time series forecasting: The forecast package for R. J. Stat. Softw. 2008, 27, 1–22. [CrossRef]
- 54. Hyndman, R.J.; Athanasopoulos, G. *Forecasting: Principles and Practice*; OTexts: Melbourne, Australia, 2018; Available online: OTexts.com/fpp2 (accessed on 15 December 2021).
- 55. McLeod, A.I.; Yu, H.; Mahdi, E. Time series analysis with R. In *Handbook of Statistics*; Elsevier: Amsterdam, The Netherlands, 2012; Volume 30, pp. 661–712.
- 56. Hyndman, R.J.; Koehler, A.B.; Snyder, R.D.; Grose, S. A state space framework for automatic forecasting using exponential smoothing methods. *Int. J. Forecast.* 2002, *18*, 439–454. [CrossRef]
- Perone, G. Comparison of ARIMA, ETS, NNAR, TBATS and hybrid models to forecast the second wave of COVID-19 hospitalizations in Italy. *Eur. J. Health Econ.* 2021, 1–24. [CrossRef] [PubMed]
- Holt, C.C. Forecasting seasonals and trends by exponentially weighted moving averages. In ONR Memorandum; PA7 Carnegie Institute of Technology Available from the Engineering Library, University of Texas at Austin: Pittsburgh, PA, USA, 1957; Volume 52.
- 59. Winters, P.R. Forecasting Sales by Exponentially Weighted Moving Averages. Manag. Sci. 1960, 6, 324–342. [CrossRef]
- De Livera, A.M.; Hyndman, R.J.; Snyder, R.D. Forecasting time series with complex seasonal patterns using exponential smoothing. J. Am. Stat. Assoc. 2011, 106, 1513–1527. [CrossRef]
- 61. Box, G.; Jenkins, G. Time Series Analysis: Forecasting and Control, 2nd ed.; Holden-Day: San Francisco, CA, USA, 1976; 537p.
- 62. Box, G.; Jenkins, G. Time Series Analysis: Forecasting and Control; Holden-Day: San Francisco, CA, USA, 1970; 575p.
- 63. Harrison, P.J.; Stevens, C.F. Bayesian Forecasting. J. R. Stat. Soc. B 1976, 38, 205–247. [CrossRef]
- 64. Abril, J.C. Structural Time Series Models. In *International Encyclopedia of Statistical Science*; Lovric, M., Ed.; Springer: Berlin/Heidelberg, Germany, 2011. [CrossRef]
- 65. Ripley, B.D. *Time series in R 1.5. 0. The Newsletter of the R Project;* R Foundation for Statistical Computing: Vienna, Austria, 2002; Volume 2, p. 2.
- 66. Pasini, A. Artificial neural networks for small dataset analysis. J. Thorac. Dis. 2015, 7, 953. [PubMed]
- 67. Allende, H.; Moraga, C.; Salas, R. Artificial neural networks in time series forecasting: A comparative analysis. *Kybernetika* **2002**, *38*, 685–707.
- Gupta, T. Deep Learning: Feedforward Neural Network. 2017. Available online: https://towardsdatascience.com/deep-learningfeedforward-neural-network-26a6705dbdc7 (accessed on 11 February 2022).
- 69. Munim, Z.H.; Shakil, M.H.; Alon, I. Next-day bitcoin price forecast. J. Risk Financ. Manag. 2019, 12, 103. [CrossRef]
- 70. Shmueli, G.; Lichtendahl, K.C., Jr. *Practical Time Series Forecasting with R: A Hands-on Guide*; Axelrod Schnall Publishers: Green Cove Springs, FL, USA, 2016.
- 71. Chai, T.; Draxler, R.R. Root Mean Square Error (RMSE) or Mean Absolute Error (MAE)?—Arguments against Avoiding RMSE in the Literature. *Geosci. Model Dev.* 2014, 7, 1247–1250. [CrossRef]
- 72. Hyndman, R.J.; Koehler, A.B. Another look at measures of forecast accuracy. Int. J. Forecast. 2006, 22, 679–688. [CrossRef]
- 73. Hassani, H.; Silva, E.S. A Kolmogorov-Smirnov Based Test for Comparing the Predictive Accuracy of Two Sets of Forecasts. *Econometrics* **2015**, *3*, 590–609. [CrossRef]
- 74. Diebold, F.X.; Mariano, R.S. Comparing predictive accuracy. J. Bus. Econ. Stat. 1995, 13, 253–263.
- 75. Harvey, D.; Leybourne, S.; Newbold, P. Testing the equality of prediction mean squared errors. *Int. J. Forecast.* **1997**, *13*, 281–291. [CrossRef]
- Diebold, F.X. Comparing Predictive Accuracy, Twenty Years Later: A Personal Perspective on the Use and Abuse of Diebold-Mariano Tests; Department of Economics, University of Pennsylvania: Philadelphia, PA, USA, 2013; pp. 1–22.
- 77. Rather, A.M.; Agarwal, A.; Sastry, V. Recurrent neural network and a hybrid model for prediction of stock returns. *Expert Syst. Appl.* **2015**, *42*, 3234–3241. [CrossRef]
- 78. United Nations Climate Change (UNFCCC). Climate Neutral Now: Guidelines for Participation. 2021. Available online: https://unfccc.int/sites/default/files/resource/CNN%20Guidelines.pdf (accessed on 5 December 2021).
- 79. Mulligan, J.; Ellison, G.; Levin, K. Foundational Questions on Carbon Removal in the United States; World Resources Institute: Washington, DC, USA, 2018.
- McKinsey. How the Voluntary Carbon Market Can Help Address Climate Change. 2020. Available online: https://www.mckinsey. com/business-functions/sustainability/our-insights/how-the-voluntary-carbon-market-can-help-address-climate-change (accessed on 21 January 2022).
- 81. Tol, R.S.J. Europe's Climate Target for 2050: An Assessment. Intereconomics 2021, 56, 330–335. [CrossRef] [PubMed]
- Dudău, R.; Cătuţi, M. The Decarbonisation Challenge of Southeast Europe: A Case Study of Romania. *Intereconomics* 2019, 54, 341–346. [CrossRef]
- 83. Moiceanu, G.; Dinca, M.N. Climate Change-Greenhouse Gas Emissions Analysis and Forecast in Romania. *Sustainability* **2021**, 13, 12186. [CrossRef]
- 84. World Bank. Romania: Climate Change and Low Carbon Green Growth Program. 2015. Available online: https://www. worldbank.org/en/country/romania/brief/romania-climate-change-and-low-carbon-green-growth-program (accessed on 22 December 2021).

- 85. Ernst and Young (EY). EY Romania Report: Renewables Can Accelerate the Decarbonisation of the Romanian Energy Sector, But Public Initiatives Must Be Synchronised with Business Intentions. 2021. Available online: https://www.ey.com/en\_ro/news/20 21/04/ey-romania-report--renewables-can-accelerate-the-decarbonisation (accessed on 22 December 2021).
- European Parliament. Climate Action in Bulgaria. 2021. Available online: https://www.europarl.europa.eu/RegData/etudes/ BRIE/2021/689330/EPRS\_BRI(2021)689330\_EN.pdf (accessed on 22 December 2021).
- Koltsaklis, N.E.; Dagoumas, A.S.; Seritan, G.; Porumb, R. Energy transition in the South East Europe: The case of the Romanian power system. *Energy Rep.* 2020, *6*, 2376–2393. [CrossRef]
- Gielen, D.; Boshell, F.; Saygin, D.; Bazilian, M.D.; Wagner, N.; Gorini, R. The role of renewable energy in the global energy transformation. *Energy Strategy Rev.* 2019, 24, 38–50. [CrossRef]
- Bilgili, F.; Koçak, E.; Bulut, Ü. The dynamic impact of renewable energy consumption on CO<sub>2</sub> emissions: A revisited Environmental Kuznets Curve approach. *Renew. Sustain. Energy Rev.* 2016, 54, 838–845. [CrossRef]
- 90. Lee, J.W. Long-run dynamics of renewable energy consumption on carbon emissions and economic growth in the European Union. *Int. J. Sustain. Dev. World Ecol.* **2019**, *26*, 69–78. [CrossRef]
- Jebli, M.B.; Farhani, S.; Guesmi, K. Renewable energy, CO<sub>2</sub> emissions and value added: Empirical evidence from countries with different income levels. *Struct. Chang. Econ. Dyn.* 2020, 53, 402–410. [CrossRef]
- 92. Koengkan, M.; Santiago, J.A.F.R. The relationship between CO<sub>2</sub> emissions, renewable and non-renewable energy consumption, economic growth, and urbanisation in the Southern Common Market. *J. Environ. Econ. Policy* **2020**, *9*, 383–401. [CrossRef]
- Usman, M.; Kousar, R.; Yaseen, M.R.; Makhdum, M.S.A. An empirical nexus between economic growth, energy utilization, trade policy, and ecological footprint: A continent-wise comparison in upper-middle-income countries. *Environ. Sci. Pollut. Res.* 2020, 27, 38995–39018. [CrossRef]
- Wolde-Rufael, Y.; Weldemeskel, E.M. Environmental policy stringency, renewable energy consumption and CO2 emissions: Panel cointegration analysis for BRIICTS countries. *Int. J. Green Energy* 2020, *17*, 568–582. [CrossRef]
- Migration Observatory. EU Migrants in Other EU Countries: An Analysis of Bilateral Migrant Stocks. 2012. Available online: https://migrationobservatory.ox.ac.uk/resources/briefings/eu-migrants-in-other-eu-countries-an-analysis-of-bilateralmigrant-stocks/ (accessed on 22 December 2021).
- Eurostat. Population Projected to Decline in Two-Thirds of EU Regions. 2021. Available online: https://ec.europa.eu/eurostat/ web/products-eurostat-news/-/ddn-20210430-2 (accessed on 22 December 2021).
- 97. Reuters. Bulgaria, Romania Rush to Offload Extra CO<sub>2</sub> Rights. 2010. Available online: https://www.reuters.com/article/uscarbon-bulgaria-romania-idUSTRE63R33I20100428 (accessed on 22 December 2021).
- 98. Clarke, L.; Jiang, K.; Akimoto, K.; Babiker, M.H.; Blanford, G.J.; Fisher-Vanden, K.A.; Hourcade, J.C.; Krey, V.; Kriegler, E.; Loeschel, A.; et al. *Assessing Transformation Pathways*; Cambridge University Press: Cambridge, UK, 2014.
- 99. Tripathi, L.; Mishra, A.K.; Dubey, A.K.; Tripathi, C.B.; Baredar, P. Renewable energy: An overview on its contribution in current energy scenario of India. *Renew. Sustain. Energy Rev.* 2016, 60, 226–233. [CrossRef]
- 100. Euractiv. A Tale of Three Countries: How Czechia, Germany, and Poland Plan to Ditch Coal. 2021. Available online: https://www. euractiv.com/section/energy/news/a-tale-of-three-countries-how-czechia-germany-and-poland-plan-to-ditch-coal/ (accessed on 22 December 2021).
- 101. Markandya, A.; Sampedro, J.; Smith, S.; Van Dingenen, R.; Pizarro-Irizar, C.; Arto, I.; González-Eguino, M. Health co-benefits from air pollution and mitigation costs of the Paris Agreement: A modelling study. *Lancet Planet. Health* 2018, 2, e126–e133. [CrossRef]
- 102. Urban, F. Climate-Change Mitigation Revisited: Low-Carbon Energy Transitions for China and India. *Dev. Policy Rev.* 2009, 27, 693–715. [CrossRef]
- 103. European Union. NextGenerationEU. 2021. Available online: https://europa.eu/next-generation-eu/index\_en (accessed on 10 December 2021).
- EUcalls. Next Generation EU: A Bold European Recovery Strategy. 2021. Available online: https://eucalls.net/blog/nextgeneration-eu (accessed on 10 December 2021).
- 105. European Council. A recovery plan for Europe. 2021. Available online: https://www.consilium.europa.eu/en/policies/eu-recovery-plan/ (accessed on 10 December 2021).
- 106. European Committee of the Regions (COTER). Integration of Geographic and Statistical Data for Better EU Policy Making. 2021. Available online: https://cor.europa.eu/en/engage/studies/Documents/CoR\_Geodata\_report.pdf (accessed on 11 February 2022).
- 107. Partnership in Statistics for Development in the 21st Century (PARIS21). Guide on Geospatial Data Integration in Official Statistics. 2021. Available online: https://paris21.org/sites/default/files/inline-files/Geospatial\_Data\_Integration\_in\_Official\_Statistics\_0. pdf (accessed on 11 February 2022).
- 108. Chief Inspectorate for Environmental Protection Database. Available online: http://powietrze.gios.gov.pl/pjp/home (accessed on 11 February 2022).
- European Commission Directorate-General for Environment. Air Quality Standards. Available online: https://ec.europa.eu/ environment/air/quality/standards.htm (accessed on 11 February 2022).
- Traczyk, P.; Gruszecka-Kosowska, A. The condition of air pollution in Kraków, Poland, in 2005–2020, with health risk assessment. *Int. J. Environ. Res. Public Health* 2020, 17, 6063. [CrossRef]

- 111. Danek, T.; Zaręba, M. The Use of Public Data from Low-Cost Sensors for the Geospatial Analysis of Air Pollution from Solid Fuel Heating during the COVID-19 Pandemic Spring Period in Krakow, Poland. *Sensors* **2021**, *21*, 5208. [CrossRef]
- 112. Airly, Intelligent Air Quality Monitoring System. Map of Air Quality by Airly. Available online: https://map.airly.org (accessed on 11 February 2022).
- 113. World Health Organization: Ambient (Outdoor) Air Quality and Health. Available online: https://www.who.int/news-room/ fact-sheets/detail/ambient-(outdoor)-air-quality-and-health (accessed on 11 February 2022).
- 114. Cohen, A.J.; Brauer, M.; Burnett, R.; Anderson, H.R.; Frostad, J.; Estep, K.; Forouzanfar, M.H. Estimates and 25-year trends of the global burden of disease attributable to ambient air pollution: An analysis of data from the Global Burden of Diseases Study 2015. *Lancet* **2017**, *389*, 1907–1918. [CrossRef]
- 115. Orellano, P.; Reynoso, J.; Quaranta, N.; Bardach, A.; Ciapponi, A. Short-term exposure to particulate matter (PM10 and PM2.5), nitrogen dioxide (NO<sub>2</sub>), and ozone (O<sub>3</sub>) and all-cause and cause-specific mortality: Systematic review and meta-analysis. *Environ. Int.* **2020**, *142*, 105876. [CrossRef]
- 116. IQAir. Air Quality in Krakow. 2022. Available online: https://www.iqair.com/poland/lesser-poland-voivodeship/krakow (accessed on 11 February 2022).